Author’s response to reviews

Title: Co-morbidities of mental disorders and chronic physical diseases in developing and emerging countries: a meta-analysis

Authors:
Labanté Daré (outcha_dare1986@yahoo.com)
Pierre-Emile Bruand (Pierre.Bruand@sanofi.com)
Daniel Gérard (d.a.gerard@wanadoo.fr)
Benoît Marin (benoit.marin@unilim.fr)
Valerie Lameyre (valerie.lameyre@sanofi.com)
Farid Boumédiène (farid.boumediene@unilim.fr)
Pierre-Marie Preux (pierre-marie.preux@unilim.fr)

Version: 4 Date: 04 Mar 2019

Author’s response to reviews:

Limoges, March 4th, 2019

To the attention of
Mrs Eva Szunyogova
BMC Public Health

Dear Editor,

Thank you very much for the opportunity to submit a revised version of our Manuscript ID PUBH-D-18-01218R3 entitled "Co-morbidities of mental disorders and chronic physical diseases in developing and emerging countries: a meta-analysis".

We thank the reviewers for their valuable comments - they helped us clarify our paper, and we have addressed all the issues.
We have attached a version of our revised paper as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

We thank you very much and look forward to hearing from you.

Sincerely,

Labanté Outcha Daré, MSc; Pierre-Emile Bruand, PharmD; Daniel Gérard, MD; Benoît Marin, PhD; Valerie Lameyre, MD; Farid Boumédiène, PhD; Pierre-Marie Preux, PhD.

INSERM, U1094, Tropical Neuroepidemiology, Limoges, France.